THE ROYAL SOCIETY PUBLISHING

PROCEEDINGS B

Where the wild things were: intrinsic and extrinsic extinction predictors in the world's most depleted mammal fauna

Samuel T. Turvey, Clare Duncan, Nathan S. Upham, Xavier Harrison and Liliana M. Dávalos

Article citation details

Proc. R. Soc. B 288: 20202905. http://dx.doi.org/10.1098/rspb.2020.2905

Review timeline

Original submission: 23 November 2020 Revised submission: Final acceptance:

11 February 2021 12 February 2021 Note: Reports are unedited and appear as submitted by the referee. The review history appears in chronological order.

Review History

RSPB-2020-2905.R0 (Original submission)

Review form: Reviewer 1

Recommendation

Accept with minor revision (please list in comments)

Scientific importance: Is the manuscript an original and important contribution to its field? Excellent

General interest: Is the paper of sufficient general interest? Excellent

Quality of the paper: Is the overall quality of the paper suitable? Excellent

Is the length of the paper justified? Yes

Should the paper be seen by a specialist statistical reviewer? No

Reports © 2021 The Reviewers; Decision Letters © 2021 The Reviewers and Editors; Responses © 2021 The Reviewers, Editors and Authors. Published by the Royal Society under the terms of the Creative Commons Attribution License http://creativecommons.org/licenses/by/4.0/, which permits unrestricted use, provided the original author and source are credited

Do you have any concerns about statistical analyses in this paper? If so, please specify them explicitly in your report.

No

It is a condition of publication that authors make their supporting data, code and materials available - either as supplementary material or hosted in an external repository. Please rate, if applicable, the supporting data on the following criteria.

Is it accessible? Yes Is it clear? Yes Is it adequate?

Yes

Do you have any ethical concerns with this paper? No

Comments to the Author

This is a rigorous and skilled analysis of extinctions of mammals species in a part of the world that has lost more species over the last 10,000 years than any other. The analysis clarifies the effects of body mass, island features and duration of human occupation on the likelihood of species extinction. This is relevant to our understanding of biodiversity change in the Caribbean, bit also to understanding global patterns of loss of mammalian species. The papers is admirably clear.

I have only one query:

At line 269-270, the authors state their decision to exclude mongoose presence as a predictor variable, because it is correlated with mean elevation (retained). This might need a bit more justification. I can understand that mean elevation may have been preferred because it is a continuous rather than a binary variable, but on the other hand I would have thought that because mongoose presence is potentially a cause of extinction it should be retained in preference to mean elevation (not itself a cause of extinction). Does this decision make any difference to results and interpretation? One of the results of the analysis is that survival is better on low-elevation islands. This could perhaps be an effect of mongoose presence, but that explanation is not really considered in the Discussion. Possibly, it would be worth testing directly whether this pattern is bet explained as a direct effect of mongoose presence/absence.

Review form: Reviewer 2

Recommendation

Accept with minor revision (please list in comments)

Scientific importance: Is the manuscript an original and important contribution to its field? Excellent

General interest: Is the paper of sufficient general interest? Good **Quality of the paper: Is the overall quality of the paper suitable?** Excellent

Is the length of the paper justified? Yes

Should the paper be seen by a specialist statistical reviewer? No

Do you have any concerns about statistical analyses in this paper? If so, please specify them explicitly in your report. N_0

It is a condition of publication that authors make their supporting data, code and materials available - either as supplementary material or hosted in an external repository. Please rate, if applicable, the supporting data on the following criteria.

Is it accessible? Yes Is it clear? Yes Is it adequate? Yes

Do you have any ethical concerns with this paper? No

Comments to the Author Where the wild things were: intrinsic and extrinsic extinction predictors in the world's most depleted mammal fauna

Comments to the author(s) Overview

The manuscript addresses the survival and extinction of mammalian species on Caribbean islands as driven by both naturally occurring and human-caused perturbations. The study focuses on survivorship and extinction selectivity on non-continental shelf islands, accounting for variation in island area, elevation, forest cover, species mass, and volcano and hurricane activity as natural drivers, but also assessing the impacts of human arrival, deforestation, and introduction of invasive species. Results show a strong selection for intermediate body mass in surviving mammals, with the extinction of larger body size being correlated to the arrival of humans. The study further underlines the importance of the increase in extreme weather events in the region, as survivorship probability is negatively associated with hurricane events. I would like to thank the authors for this well-written and interesting manuscript. Overall, I found it to be an insightful and the discussion to be well and thoroughly thought out. The consideration of multiple time events in human driven perturbations in comparison the natural environmental drivers was particularly interesting. I appreciate the conclusions highlighting particular island regions that could be concentrated on in the future for conservation efforts, making the results of this more applicable to modern ecosystems. I have a few suggestions for consideration which I hope will help strengthen certain points made here.

General Comments

Elevation and invasive species introductions: Due to finding strong collinearity between mean elevation, maximum elevation, and mongoose presence the authors chose to exclude both maximum elevation and mongoose presence from further analysis. While I understand the removal based on collinearity within the analyses, the removal of mongoose (which here I understood to be a major representative of an invasive species with dramatic effects to natural populations) should be expanded upon in the discussion. While the authors do have some discussion on invasive species, because of this removal the focus is drawn to mean elevation, which may diminish the importance of biotic interactions between species. Mean island elevation was negatively correlated with survival probability, so I would assume mongoose presence would be as well, but the discussion may benefit from taking a deeper dive into this potential overlap of variables. What does this mean when they are considered together?

Mass and selectivity for intermediate size: The strong selectivity for intermediate size in these island populations is highlighted in this study. The authors do a good job discussing how they think each variable may be affecting lower and higher mass groups differently, with humans appearing to negatively impact upper mass groups of mammals and invasive species likely accounting for the removal of lower mass groups. The discussion may benefit from a discussion of how the compression towards an intermediate size may have compounding effects on survivorship or extinction in the future. This may also benefit the concluding remarks highlighting the conservation of certain Caribbean ecosystems for the surviving species of the region.

Additionally, as well as the comparison to Australian fauna, consider adding a couple specific examples to other oceanic islands of more similar sizes to that of your study for further clarity (you may check out Heaney et al 1989 and Alcover et al 1998).

Figures and Tables: It seems that Figure 2 and Table 2 give the same information. I therefore do not think it is necessary to provide both as it does not provide the reader with too much extra information to interpret the results of the study. It may therefore make sense to move one or the other to the supplementary materials section. I personally find the figure easier to compare across the sample-wide effects, though I recognize that the extremes of the high probability intervals for "active volcano" and "forest loss" are not visible in this graphic.

References

Alcover JA, Sans A, Palmer M. 1998 The extent of extinctions of mammals on islands. J. Biogeography 25, 913-918.

Heaney, LR, Heideman PD, Rickart EA, Utzurrum RB, Klompen JSH. 1989 Elevational zonation of mammals in the central Philipiines. J. Trop. Ecol. 5, 259-280.

Decision letter (RSPB-2020-2905.R0)

05-Feb-2021

Dear Dr Turvey

I am pleased to inform you that your manuscript RSPB-2020-2905 entitled "Where the wild things were: intrinsic and extrinsic extinction predictors in the world's most depleted mammal fauna" has been accepted for publication in Proceedings B.

The referee(s) have recommended publication, but also suggest some minor revisions to your manuscript. Therefore, I invite you to respond to the referee(s)' comments and revise your

manuscript. Because the schedule for publication is very tight, it is a condition of publication that you submit the revised version of your manuscript within 7 days. If you do not think you will be able to meet this date please let us know.

To revise your manuscript, log into https://mc.manuscriptcentral.com/prsb and enter your Author Centre, where you will find your manuscript title listed under "Manuscripts with Decisions." Under "Actions," click on "Create a Revision." Your manuscript number has been appended to denote a revision. You will be unable to make your revisions on the originally submitted version of the manuscript. Instead, revise your manuscript and upload a new version through your Author Centre.

When submitting your revised manuscript, you will be able to respond to the comments made by the referee(s) and upload a file "Response to Referees". You can use this to document any changes you make to the original manuscript. We require a copy of the manuscript with revisions made since the previous version marked as 'tracked changes' to be included in the 'response to referees' document.

Before uploading your revised files please make sure that you have:

1) A text file of the manuscript (doc, txt, rtf or tex), including the references, tables (including captions) and figure captions. Please remove any tracked changes from the text before submission. PDF files are not an accepted format for the "Main Document".

2) A separate electronic file of each figure (tiff, EPS or print-quality PDF preferred). The format should be produced directly from original creation package, or original software format. PowerPoint files are not accepted.

3) Electronic supplementary material: this should be contained in a separate file and where possible, all ESM should be combined into a single file. All supplementary materials accompanying an accepted article will be treated as in their final form. They will be published alongside the paper on the journal website and posted on the online figshare repository. Files on figshare will be made available approximately one week before the accompanying article so that the supplementary material can be attributed a unique DOI.

Online supplementary material will also carry the title and description provided during submission, so please ensure these are accurate and informative. Note that the Royal Society will not edit or typeset supplementary material and it will be hosted as provided. Please ensure that the supplementary material includes the paper details (authors, title, journal name, article DOI). Your article DOI will be 10.1098/rspb.[paper ID in form xxxx.xxxx e.g. 10.1098/rspb.2016.0049].

4) A media summary: a short non-technical summary (up to 100 words) of the key findings/importance of your manuscript.

5) Data accessibility section and data citation

It is a condition of publication that data supporting your paper are made available either in the electronic supplementary material or through an appropriate repository. Please see our Data Sharing Policies https://royalsociety.org/journals/authors/author-guidelines/#data.

In order to ensure effective and robust dissemination and appropriate credit to authors the dataset(s) used should be fully cited. To ensure archived data are available to readers, authors should include a 'data accessibility' section immediately after the acknowledgements section. This should list the database and accession number for all data from the article that has been made publicly available, for instance:

- DNA sequences: Genbank accessions F234391-F234402
- Phylogenetic data: TreeBASE accession number S9123
- Final DNA sequence assembly uploaded as online supplemental material

• Climate data and MaxEnt input files: Dryad doi:10.5521/dryad.12311

NB. From April 1 2013, peer reviewed articles based on research funded wholly or partly by RCUK must include, if applicable, a statement on how the underlying research materials – such as data, samples or models – can be accessed. This statement should be included in the data accessibility section.

If you wish to submit your data to Dryad (http://datadryad.org/) and have not already done so you can submit your data via this link

http://datadryad.org/submit?journalID=RSPB&manu=(Document not available) which will take you to your unique entry in the Dryad repository. If you have already submitted your data to dryad you can make any necessary revisions to your dataset by following the above link. Please see https://royalsociety.org/journals/ethics-policies/data-sharing-mining/ for more details.

6) For more information on our Licence to Publish, Open Access, Cover images and Media summaries, please visit https://royalsociety.org/journals/authors/author-guidelines/.

Once again, thank you for submitting your manuscript to Proceedings B and I look forward to receiving your revision. If you have any questions at all, please do not hesitate to get in touch.

Sincerely, Dr Sasha Dall mailto: proceedingsb@royalsociety.org

Associate Editor Board Member: 1 Comments to Author:

I agree with both reviewers that this manuscript is well-written, rigorous and an important question of broad interest. Both reviewers had the same minor concern about omitting mongoose as a predictor variable rather than another correlated variable, given the important role of this invasive predator as a known mechanism of mammal loss on Caribbean islands. I agree that this needs justification.

Reviewer(s)' Comments to Author:

Referee: 1

Comments to the Author(s)

This is a rigorous and skilled analysis of extinctions of mammals species in a part of the world that has lost more species over the last 10,000 years than any other. The analysis clarifies the effects of body mass, island features and duration of human occupation on the likelihood of species extinction. This is relevant to our understanding of biodiversity change in the Caribbean, bit also to understanding global patterns of loss of mammalian species. The papers is admirably clear.

I have only one query:

At line 269-270, the authors state their decision to exclude mongoose presence as a predictor variable, because it is correlated with mean elevation (retained). This might need a bit more justification. I can understand that mean elevation may have been preferred because it is a continuous rather than a binary variable, but on the other hand I would have thought that because mongoose presence is potentially a cause of extinction it should be retained in preference to mean elevation (not itself a cause of extinction). Does this decision make any difference to results and interpretation? One of the results of the analysis is that survival is better on low-elevation islands. This could perhaps be an effect of mongoose presence, but that explanation is not really considered in the Discussion. Possibly, it would be worth testing directly whether this pattern is bet explained as a direct effect of mongoose presence/absence.

Referee: 2

Comments to the Author(s) Where the wild things were: intrinsic and extrinsic extinction predictors in the world's most depleted mammal fauna

Comments to the author(s)

Overview

The manuscript addresses the survival and extinction of mammalian species on Caribbean islands as driven by both naturally occurring and human-caused perturbations. The study focuses on survivorship and extinction selectivity on non-continental shelf islands, accounting for variation in island area, elevation, forest cover, species mass, and volcano and hurricane activity as natural drivers, but also assessing the impacts of human arrival, deforestation, and introduction of invasive species. Results show a strong selection for intermediate body mass in surviving mammals, with the extinction of larger body size being correlated to the arrival of humans. The study further underlines the importance of the increase in extreme weather events in the region, as survivorship probability is negatively associated with hurricane events. I would like to thank the authors for this well-written and interesting manuscript. Overall, I found it to be an insightful and the discussion to be well and thoroughly thought out. The consideration of multiple time events in human driven perturbations in comparison the natural environmental drivers was particularly interesting. I appreciate the conclusions highlighting particular island regions that could be concentrated on in the future for conservation efforts, making the results of this more applicable to modern ecosystems. I have a few suggestions for consideration which I hope will help strengthen certain points made here.

General Comments

Elevation and invasive species introductions: Due to finding strong collinearity between mean elevation, maximum elevation, and mongoose presence the authors chose to exclude both maximum elevation and mongoose presence from further analysis. While I understand the removal based on collinearity within the analyses, the removal of mongoose (which here I understood to be a major representative of an invasive species with dramatic effects to natural populations) should be expanded upon in the discussion. While the authors do have some discussion on invasive species, because of this removal the focus is drawn to mean elevation, which may diminish the importance of biotic interactions between species. Mean island elevation was negatively correlated with survival probability, so I would assume mongoose presence would be as well, but the discussion may benefit from taking a deeper dive into this potential overlap of variables. What does this mean when they are considered together?

Mass and selectivity for intermediate size: The strong selectivity for intermediate size in these island populations is highlighted in this study. The authors do a good job discussing how they think each variable may be affecting lower and higher mass groups differently, with humans appearing to negatively impact upper mass groups of mammals and invasive species likely accounting for the removal of lower mass groups. The discussion may benefit from a discussion of how the compression towards an intermediate size may have compounding effects on survivorship or extinction in the future. This may also benefit the concluding remarks highlighting the conservation of certain Caribbean ecosystems for the surviving species of the region.

Additionally, as well as the comparison to Australian fauna, consider adding a couple specific examples to other oceanic islands of more similar sizes to that of your study for further clarity (you may check out Heaney et al 1989 and Alcover et al 1998).

Figures and Tables: It seems that Figure 2 and Table 2 give the same information. I therefore do not think it is necessary to provide both as it does not provide the reader with too much extra information to interpret the results of the study. It may therefore make sense to move one or the other to the supplementary materials section. I personally find the figure easier to compare across the sample-wide effects, though I recognize that the extremes of the high probability intervals for "active volcano" and "forest loss" are not visible in this graphic.

References

Alcover JA, Sans A, Palmer M. 1998 The extent of extinctions of mammals on islands. J. Biogeography 25, 913-918.

Heaney, LR, Heideman PD, Rickart EA, Utzurrum RB, Klompen JSH. 1989 Elevational zonation of mammals in the central Philipiines. J. Trop. Ecol. 5, 259-280.

Decision letter (RSPB-2020-2905.R1)

12-Feb-2021

Dear Dr Turvey

I am pleased to inform you that your manuscript entitled "Where the wild things were: intrinsic and extrinsic extinction predictors in the world's most depleted mammal fauna" has been accepted for publication in Proceedings B.

You can expect to receive a proof of your article from our Production office in due course, please check your spam filter if you do not receive it. PLEASE NOTE: you will be given the exact page length of your paper which may be different from the estimation from Editorial and you may be asked to reduce your paper if it goes over the 10 page limit.

If you are likely to be away from e-mail contact please let us know. Due to rapid publication and an extremely tight schedule, if comments are not received, we may publish the paper as it stands.

If you have any queries regarding the production of your final article or the publication date please contact procb_proofs@royalsociety.org

Your article has been estimated as being 8 pages long. Our Production Office will be able to confirm the exact length at proof stage.

Open Access

You are invited to opt for Open Access, making your freely available to all as soon as it is ready for publication under a CCBY licence. Our article processing charge for Open Access is £1700. Corresponding authors from member institutions

(http://royalsocietypublishing.org/site/librarians/allmembers.xhtml) receive a 25% discount to these charges. For more information please visit http://royalsocietypublishing.org/open-access.

Paper charges

An e-mail request for payment of any related charges will be sent out shortly. The preferred payment method is by credit card; however, other payment options are available.

Electronic supplementary material:

All supplementary materials accompanying an accepted article will be treated as in their final form. They will be published alongside the paper on the journal website and posted on the online figshare repository. Files on figshare will be made available approximately one week before the accompanying article so that the supplementary material can be attributed a unique DOI.

You are allowed to post any version of your manuscript on a personal website, repository or preprint server. However, the work remains under media embargo and you should not discuss it with the press until the date of publication. Please visit https://royalsociety.org/journals/ethics-policies/media-embargo for more information.

Thank you for your fine contribution. On behalf of the Editors of the Proceedings B, we look forward to your continued contributions to the Journal.

Sincerely, Editor, Proceedings B mailto: proceedingsb@royalsociety.org